PLOS NEGLECTED TROPICAL DISEASES



GOPEN ACCESS

Citation: Ximenes RAdA, Miranda-Filho DdB, Montarroyos UR, Martelli CMT, Araújo TVBd, Brickley E, et al. (2021) Zika-related adverse outcomes in a cohort of pregnant women with rash in Pernambuco, Brazil. PLoS Negl Trop Dis 15(3): e0009216. https://doi.org/10.1371/journal. pntd.0009216

Editor: Ran Wang, Beijing Children's Hospital, Capital Medical University, CHINA

Received: August 24, 2020

Accepted: February 7, 2021

Published: March 8, 2021

Copyright: © 2021 Ximenes et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: Data cannot be shared publicly because public availability would compromise patient privacy. De-identified data can be made available upon reasonable request from qualified investigators by contacting the Programa de Pós-Graduação em Ciências da Saúde (PPGCS) da Universidade de Pernambuco (UPE) at ppg. cienciasdasaude@upe.br.

Funding: This work was supported by the: European Union's Horizon 2020 Research and RESEARCH ARTICLE

Zika-related adverse outcomes in a cohort of pregnant women with rash in Pernambuco, Brazil

Ricardo Arraes de Alencar Ximenes^{1,2*}, Demócrito de Barros Miranda-Filho², Ulisses Ramos Montarroyos³, Celina Maria Turchi Martelli⁴, Thalia Velho Barreto de Araújo⁵, Elizabeth Brickley⁶, Maria de Fátima Pessoa Militão de Albuquerque⁴, Wayner Vieira Souza⁴, Liana O. Ventura⁷, Camila V. Ventura⁷, Adriana L. Gois⁷, Mariana C. Leal⁸, Danielle Maria da Silva Oliveira⁹, Sophie Helena Eickmann¹⁰, Maria Durce C. G. Carvalho¹¹, Paula F. S. da Silva¹¹, Maria Angela Wanderley Rocha², Regina Coeli Ferreira Ramos¹², Sinval Pinto Brandão-Filho¹³, Marli Tenorio Cordeiro¹⁴, Luciana Caroline Albuquerque Bezerra¹⁵, George Dimech¹⁵, Sandra Valongueiro⁵, Pedro Pires¹⁶, Priscila Mayrelle da Silva Castanha¹⁷, Rafael Dhalia¹⁴, Ernesto Torres Azevedo Marques-Júnior¹⁴, Laura C. Rodrigues⁶, on behalf of the Microcephaly Epidemic Research Group (MERG)

1 Departamento de Medicina Tropical, Universidade Federal de Pernambuco, Recife, Pernambuco, Brazil, 2 Faculdade de Ciências Médicas, Universidade de Pernambuco, Recife, Pernambuco, Brazil, 3 Instituto de Ciências Biológicas, Universidade de Pernambuco, Recife, Pernambuco, Brazil, 4 Departamento de Saúde Coletiva, Instituto Aggeu Magalhães, Recife, Pernambuco, Brazil, 5 Departamento de Medicina Social, Universidade Federal de Pernambuco, Recife, Pernambuco, Brazil, 6 Department of Infectious Disease Epidemiology, London School of Hygiene & Tropical Medicine, London, United Kingdom, 7 Departamento de Oftalmologia, Fundação Altino Ventura, Recife, Pernambuco, Brasil, 8 Departamento de Cirurgia, Universidade Federal de Pernambuco, Recife, Pernambuco Brazil, 9 UTI Pediátrica, Hospital Universitário Oswaldo Cruz, Recife, Pernambuco, Brazil, 10 Departamento Materno-Infantil, Universidade Federal de Pernambuco, Recife, Pernambuco, Brazil, 11 Departamento de Neurologia, Hospital Universitário Oswaldo Cruz, Recife, Pernambuco, Brazil, 12 Departamento de Infectologia Pediátrica, Hospital Universitário Oswaldo Cruz, Recife, Pernambuco, Brazil, 13 Departamento de Imunologia, Instituto Aggeu Maglhães, Recife, Pernambuco, Brazil, 14 Laboratório de Virologia e Terapia Experimental, Instituto Aggeu Maglhães, Recife, Pernambuco, Brazil. 15 Secretaria Executiva da Vigilância em Saúde, Secretaria de Saúde de Pernambuco, Recife, Pernambuco, Brazil, 16 Departamento Materno-Infantil, Universidade de Pernambuco, Recife, Pernambuco, Brazil, 17 Programa de Pós-graduação em Ciências da Saúde, Universidade de Pernambuco, Recife, Pernambuco, Brazil

* raaximenes@uol.com.br

Abstract

Background

While Zika virus (ZIKV) is now widely recognized as a teratogen, the frequency and full spectrum of adverse outcomes of congenital ZIKV infection remains incompletely understood.

Methods

Participants in the MERG cohort of pregnant women with rash, recruited from the surveillance system from December/2015-June/2017. Exposure definition was based on a combination of longitudinal data from molecular, serologic (IgM and IgG3) and plaque reduction neutralization tests for ZIKV. Children were evaluated by a team of clinical specialists and

Innovation programme (https://ec.europa.eu/ programmes/horizon2020/) under ZikaPLAN grant agreement (No. 734584 to RAAX), (https:// zikaplan.tghn.org/); Wellcome Trust & the UK Department for International Development (201870/Z/16/Z to DBMF & 205377/Z/16/Z to RAAX, https://wellcome.ac.uk/); Wellcome Trust -**Research Enrichment in Epidemic Situations** (107779/Z/15/Z, with ER1505 & ER1601 to RAAX); Medical Research Council on behalf of the Newton Fund and Wellcome Trust (MC PC 15088; https:// mrc.ukri.org/); Fundação de Amparo à Ciência e Tecnologia de PE - FACEPE (APQ-0192-4.01/17 to Programa de Pós-graduação em Ciências da Saúde-UPE and APQ-0172-4.01/16 to DBMF, http://www.facepe.br/); Coordenação de Aperfeicoamento de Pessoal de Nível Superior -Brasil (CAPES) - Finance Code 001; Scholarship: Conselho Nacional de Desenvolvimento Científico e Tecnológico (309722/2017-9 to RAAX, 306708/ 2014-0 to CMTM; http://www.cnpq.br/), Secretaria de Vigilância em Saúde/ Ministério da Saúde de Brasil Resposta à Emergência em Saúde Pública -Zika vírus e Microcefalia (837058/2016 to RAAX); Conselho Nacional de Desenvolvimento Científico e Tecnológico, Coordenação de Aperfeiçoamento de Pessoal de Nível Superior (http://www.capes.gov. br/); Departamento de Ciência e Tecnologia (http:// rebrats.saude.gov.br/noticias/59DECIT) for Prevenção e Combate ao vírus Zika I (440839/ 2016-5 to RAAX). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

by transfontanelle ultrasound and were classified as having microcephaly and/or other signs/symptoms consistent with congenital Zika syndrome (CZS). Risks of adverse outcomes were quantified according to the relative evidence of a ZIKV infection in pregnancy.

Findings

376 women had confirmed and suspected exposure to ZIKV. Among evaluable children born to these mothers, 20% presented with an adverse outcome compatible with exposure to ZIKV during pregnancy. The absolute risk of microcephaly was 2.9% (11/376), of calcifications and/or ventriculomegaly was 7.2% (13/180), of additional neurologic alterations was 5.3% (13/245), of ophthalmologic abnormalities was 7% (15/214), and of dysphagia was 1.8% (4/226). Less than 1% of the children experienced abnormalities across all of the domains simultaneously. Interpretation: Although approximately one-fifth of children with confirmed and suspected exposure to ZIKV in pregnancy presented with at least one abnormality compatible with CZS, the manifestations presented more frequently in isolation than in combination. Due to the rare nature of some outcomes and the possibility of later manifestations, large scale individual participant data meta-analysis and the long-term evaluation of children are imperative to identify the full spectrum of this syndrome and to plan actions to reduce damages.

Author summary

The ability to meaningfully quantify the absolute and relative risks of Congenital Zika Syndrome is contingent on the accurate identification of ZIKV infections in pregnant women and the long-term follow-up of children at risk. This study builds on the evidence base on ZIKV in pregnancy by not only examining pregnancy and birth outcomes, but also considering later onset manifestations of Congenital Zika Syndrome, conducting a deep and standardized investigation of infant outcomes. In addition, it used unprecedented repeated testing and the use of multiple diagnostic platforms, including qRT-PCR, IgM and IgG3 ELISAs, and PRNTs to identify ZIKV cases. The absolute risk of microcephaly was 2.9%, of calcifications and/or ventriculomegaly was 7.2%, of additional neurologic alterations was 5.3%, of ophthalmologic abnormalities was 7% and of dysphagia was 1.8%. The manifestations presented more frequently in isolation than in combination (i.e., less than 1% of the children experienced abnormalities across all of the domains simultaneously). Although approximately one-fifth of children with confirmed and suspected exposure to ZIKV in pregnancy presented with at least one abnormality compatible with CZS, the manifestations presented more frequently in isolation than in combination

Introduction

Since the start of the microcephaly epidemic in the Americas in 2015, evidence detailing the consequences of prenatal exposure to Zika virus (ZIKV) has been accumulating. Case reports provided vital descriptions mainly of affected children with severe manifestations [1], and shed light on both common features (e.g., calcifications [2]) as well as organ system-specific signs and symptoms (e.g., cryptorchidism [3]). A constellation of structural birth defects and

functional disabilities is now clinically recognizable as Congenital Zika Syndrome (CZS) [4,5], though there are no standard criteria for diagnosing CZS. Key questions persist related to the relative frequency of CZS and its overlapping clinical features.

To date, several cohort studies of ZIKV-infected pregnant women have begun to estimate the risks of adverse pregnancy and birth outcomes after infection [6–13] and consistently estimate the absolute risk of microcephaly to be approximately 5%. However, there is wide variation in the estimated frequencies of other adverse outcomes, ranging from 6 [7] to 46% [6]. These other outcomes mainly include grossly abnormal clinical or brain imaging findings or both [6], often accompanied by neurologic and ocular defects [7]. As methodological issues limit the comparability of the existing findings, the spectrum of abnormalities arising from prenatal exposure to ZIKV and the risk of each of them—either separately or in combination —remains incompletely understood.

Pernambuco in the northeast of Brazil was the epicenter of the microcephaly epidemic that, along with a cluster of other neurological disorders, triggered the World Health Organization's declaration of a Public Health Emergency of International Concern[14]. At the request of the Brazilian Ministry of Health and the State Health Secretariat, the Microcephaly Epidemic Research Group (MERG) was assembled, bringing together multidisciplinary investigators from several institutions across Pernambuco state and abroad. The MERG was tasked with providing an initial description of microcephaly using a case-control study [15,16], and estimating the risks of adverse outcomes using a cohort study of pregnant women.

The pregnancy cohort study, MERG cohort, was designed at the height of the epidemic in coordination with the public health response efforts of the Pernambuco State Health Secretariat. This study adopted a unique approach evaluating the children by different groups of specialists using standardized protocol and forms, on the same day, and quantifying the risks according to the relative evidence for a ZIKV infection in pregnancy, contributing to understanding of the risks and spectrum of CZS. This comprehensive evaluation allowed us to investigate a range of adverse outcomes compatible with the exposure to ZIKV infection during pregnancy.

Methods

Ethics statement

The project was approved by the Research Ethics Committee of the Aggeu Magalhães / Fiocruz Research Center, with CAAE: 53240816.4.0000.5190 and followed the ethical procedures recommended for this type of study by Resolution MS/CNS 466/2012. Free and informed written consent was requested for all mothers/legal guardians of children.

Study design and data collection

Following the identification of the microcephaly epidemic, in December 2015, the Pernambuco State Health Department introduced a surveillance system for pregnant women presenting with rash (Center for Strategic Information on Health Surveillance in Pernambuco; CIEVS/PE). Participants in the MERG cohort were recruited from Pernambuco state's compulsory notification system for pregnant women with rash. The surveillance protocols of the Pernambuco State Health Secretariat and MERG protocols were harmonized, and some activities shared. Pregnant women with rash were notified to the surveillance program by the health units, and blood samples collected within five days of rash onset (time point 1). At least 14 days after notification, the MERG fieldworkers recruited these women into this study, collected a second blood sample and administered a standardized questionnaire (time point 2). In cases of livebirths, a third blood sample was collected from participating women after delivery (time point 3). Testing at each timepoint was independent of the others (i.e., women could be tested at one point even if they had not been tested in the previous one).

Recruitment commenced in December/2015 and included women residing up to 120km from the Metropolitan Region of Recife. The first blood samples were sent to the Central Laboratory of Public Health in Recife, where were separated and stored at -80°C until further diagnostic testing was performed at the Laboratorio de Virologia e Terapia Experimental of the Fundação Oswaldo Cruz (LAVITE-Fiocruz). The second and third blood samples were separated, stored, and tested at LAVITE-Fiocruz.

At baseline, pregnant women were evaluated in terms of sociodemographic factors (age, race/ethnicity, years of education, social class, and monthly per capita family income), reproductive history (previous pregnancy, children with malformations from previous pregnancies), current pregnancy characteristics (smoking, illicit drug use, delivery mode). For social class we used the new Brazilian Economic Classification Criteria [17], which is based on the pattern of consumption of the Brazilian families. It ranges from A (higher) to E (lower). Smoking and illicit drug use were defined according to the information given by the pregnant women in response to the questionnaire, independently of the gestational age and of the frequency and quantity.

Gestational age at the time of rash onset and delivery was estimated, in order of priority, by first trimester ultrasound, the date of last menstrual period, and any other trimester ultrasound.

Diagnostic assays

The main exposure, prenatal ZIKV infection, was based on clinical (rash in any form), and laboratory criteria, based on the combination of longitudinal data from nucleic acid amplification tests, serologic assays and plaque reduction neutralization tests, as previously described [18]. Further laboratory tests were performed in the pregnant women with rash to test for TORCH agents and other arboviruses.

Molecular assay. Maternal sera were tested for the detection of ZIKV RNA by one-step qRT-PCR using primers and probes previously described [19].

Serologic assays. Samples were screened for the detection of ZIKV-specific IgM antibodies by capture-IgM enzyme-linked immunosorbent assay (ELISA) [20]. Maternal sera were also tested for the detection of ZIKV-specific IgG3 anti-non-structural protein 1 (NS1) antibodies using a novel in-house ELISA following a protocol previously described [21]. ZIKVspecific neutralizing activity was assessed in all available maternal sera by PRNT, following a standardized protocol [22] carried out in Vero cells using a virus strain isolated in the study setting: ZIKV (strain Brazil/PE243/2015) [23]. The cut-off value for PRNT positivity was defined based on a 50% reduction in plaque counts (PRNT50). ZIKV-specific neutralizing antibody titers were estimated by four-parameter nonlinear regression using the sigmoidal dose response (variable slope) equation on GraphPad Prism 7.0a. Maternal sera were considered ZIKV-non-negative with PRNT50 titers \geq 20, equivocal with PRNT50 titers \geq 20 and <100, and ZIKV-positive with PRNT50 titers \geq 100. Seroconversion was considered to occur with four-fold rises in PRNT50 titers or with a switch from negative (i.e., <20) to non-negative status (i.e., \geq 20). Maternal sera of a subsample was tested for DENV using Panbio Dengue IgG Indirect ELISA.

Degrees of evidence of ZIKV infection

We categorized maternal ZIKV exposure according to the degree of evidence of infection (See Ximenes et al. 2019 [18] for detailed description). Pregnant women were classified as Positive

for ZIKV infection, Suspected or Negative. The positive group was divided into three subcategories (robust, moderate and limited evidence) according to the level of diagnostic evidence. Robust evidence was defined by a positive qRT-PCR, seroconversion, or at least two positive serologic tests in pregnancy or by one positive IgM or IgG3 in pregnancy paired with a nonnegative PRNT50 within six months post-pregnancy. Moderate evidence was defined by one positive IgM or IgG3 in pregnancy, an indication of seroconversion by PRNT50 during pregnancy (i.e., either a PRNT50 titer \geq 1000 in pregnancy paired with a rise within 2 months postpregnancy or 4-fold rise in PRNT50 titer from pregnancy to within 2 months post-pregnancy), or an equivocal PRNT50 test result in pregnancy paired with a positive PRNT50 within three months post-pregnancy. Limited evidence of ZIKV infection was defined by a positive PRNT50 in pregnancy or within 6 months post-pregnancy or an indication of PRNT50 seroconversion during the 2 to 3 months post-pregnancy. The Suspected ZIKV infection group included two subcategories (limited evidence of flavivirus and inconclusive results). Limited evidence of a flavivirus before or during pregnancy was defined by a PRNT50 titer between 20 and 100 or a non-negative result (i.e., unspecified titer \geq 20) in pregnancy or within 1 month post-pregnancy, providing evidence of infection by an unspecified flavivirus. The inconclusive subgroup included pregnant women who tested IgM negative for ZIKV, but were tested more than three months after the rash started, and women who showed inconsistency in the results of the same test. We considered there to be evidence against ZIKV infection if all tests performed in pregnancy were negative.

Pediatric assessment

All children born to the mothers of the MERG Pregnant Women Cohort were referred to just one health unit where they were evaluated by different specialists of the MERG team, using standardized protocols especially created for these assessments. Transfontanelle ultrasound (TF-US) was performed within the first six months of life. Priority was given to those children born to mothers with lab evidence of ZIKV infection, independently of the level of evidence, but the evaluation was carried out for all children independently of suspicious of an abnormality. The timing of evaluation was constrained by the operational capacity of the research team, as the Northeast of Brazil was the place where the cases of microcephaly were first identified and Pernambuco was in the epicentre of the epidemic, and the study was designed and conducted when the epidemic was already occurring. This evaluation of the outcomes of the pregnant women cohort was also the baseline evaluation of the MERG Paediatric Cohort.

Pediatricians evaluated infants for malformations or other abnormalities. Microcephaly was defined as a head circumference falling two or more standard deviations below the mean for age and sex. For infants born with gestational age \geq 37 weeks, head circumference was evaluated using the WHO Anthro software and classified according to the World Health Organization (2007) growth charts [24]. For premature infants, age was corrected for gestational age at delivery and classified using the Intergrowth (2015) growth charts [25]. Cases were classified as having disproportionate microcephaly based on the ratio of head circumference to length [2]. Where head circumference in cm was less than ½ length in cm –7 (1/2L-7), microcephaly cases were considered to be disproportionate.

Pediatric neurologists examined children for: seizures, altered consciousness/behavior, localized motor deficits, altered tonus or trophism, and pyramidal signs. Neuroimaging results from the TF-US were assessed for: calcifications, cerebellar hypoplasia or atrophy, agenesis or dysgenesis of the corpus callosum and diffuse cortical atrophy.

Otolaryngologists performed clinical and auditory evaluations, including the short latency auditory brainstem response to stimuli. If the first screening test was abnormal, the test was

repeated later. If the second test also indicated hearing loss, the child was referred for further diagnosis.

Speech therapists evaluated infants' ability to swallow and the stomatognathic system, following a clinical protocol based on the Pediatric Dysphagia Assessment Protocol [26].

Pediatric ophthalmologists and retina specialists from the Altino Ventura Foundation, evaluated children for: visual acuity testing, oculomotility testing, cycloplegic refraction, indirect ophthalmoscopy, and wide-angle fundus imaging (Retcam Shuttle, Clarity Medical Systems, Pleasanton, California, United States of America).

Children were classified as having signs and symptoms consistent with CZS if they had one or more of the following: microcephaly, neuroimaging, neurologic, auditory and ophthalmologic abnormalities.

Statistical analysis

We estimated the absolute risk and respective 95% confidence interval of microcephaly and other adverse outcomes potentially related to prenatal exposure to Zika virus infection separately for each outcome and for the combined presentation. The analysis was stratified according to the level of laboratory evidence, allowing a sensitivity analyses. We also estimated the relative risk, 95% confidence interval and p-value (Chi-squared test) for the association between trimester of gestation in which the infection occurred, years of education and social class with microcephaly and with any of the other abnormalities. For comparing the clinical features of ZIKV infection we used the Chi-squared test or Fisher's exact tests for categorical variables and the Mann-Whitney U tests for continuous variables. We used the Stata software (version 14.1 (College Station, Texas, USA)) for the statistical analyses.

Results

A total of 694 pregnant women with rash was recruited between December/2015 and June/ 2017. Out of the total, 364 (52.4%) women were tested for ZIKV infection after delivery (at time point 3), 240 (34.6%) at 14 or more days after notification (at time point 2) and 90 (13.0%) within five days of rash onset (at time point 1). A total of 503 children born to these women were evaluated sometime after birth (Fig 1). In relation to the exposure status of their mothers, 277 were classified as having some level of laboratory evidence of ZIKV infection during pregnancy, 78 of unspecified Flavivirus and for 19 the results were inconclusive. For 127 there was no laboratory evidence of ZIKV infection (Table 1).

The time elapsed between birth and first evaluation had a median of 4.65 months (IQR: 2.9–13.4). There were two foetal deaths, one with a gestational age of approximately 5 months and the other with a gestational age of 34 weeks. Five deaths occurred after delivery, two within one day, one within two months and two after a year. The two latter deaths occurred in children with Down Syndrome. For three of these five children, including those with Down Syndrome, there was a report of severe cardiac malformation.

The sociodemographic characteristics, reproductive history and substance use of the pregnant women are presented in Table 1 and S1 Table.

Rash was referred in the interview by 98% of the women. Approximately 80% informed having had fever, joint pain and fatigue/feebleness. Joint pain, joint swelling and presence of secretion in the eyes when wake up were more frequent among women who had laboratory evidence of ZIKV infection (Table 2).

Table 3 shows the absolute risk of the main adverse outcomes according to the level of ZIKV exposure during pregnancy. The columns in bold summarise the information for all positive categories (i.e., robust, moderate, and limited evidence) and the combined positive



^a The dyads include 501 women and 503 children, including two twin pairs



https://doi.org/10.1371/journal.pntd.0009216.g001

plus suspected categories (i.e., robust, moderate, and limited evidence plus flavivirus and inconclusive). In addition to absolute risks at first evaluation, we also added prospective information of the risk of presenting with neurologic and ophthalmologic abnormalities and dysphagia at any time during the follow up. Of the 503 pregnant women with rash, 12 cases of microcephaly were identified, one occurring in a child born to a woman with no laboratory evidence of Zika virus infection. Among the 376 women who were ZIKV positive or had suspected ZIKV exposure, the absolute risk of microcephaly was approximately 3% (Table 3). Among these cases, three were disproportionate. Nine had a TF-US performed and no abnormality was observed in 7; in one child cranial haemorrhage grade I was observed and in another, born to the woman with no laboratory evidence of ZIKV infection, alteration of the optic nerve and retina, a benign lump of the subaracnoid space and asymmetry were described. For two children (out of three) with disproportionate microcephaly and TF-US performed, no abnormality was found.

Of 263/503 children who had TF-US performed, 30 had some imaging alteration, of which 21 were among non-negative. The absolute risk of any imaging alteration in children born to non-negative mothers was 9.2% and of calcifications or ventriculomegaly 5.7%, but these risks were similar among the offspring of negative mothers. Of the five children who had imaging alteration and were born to ZIKV negative mothers, two were born to Chikungunya positive mothers, one being PCR positive and, the other being IgM positive; two were born to mothers negative for TORCH, Dengue and Chikungunya and one to a mother who was not tested for TORCH.

Sociodemographic factors Image: Parts, Median (UQR) 501 26.2 (21-31) Race/ethnicity 500	Variables	Total ^a	N (%)
Age, years, Median (IQR) 501 262 (21-31) Racclethnicity 500 "Preto" (ic, black) 501 "Pardo" (ic, mixed race) 335 (670) "Branco" (ic, white) 107 (21.4) Other 8 (1.6) Years of education 501 0-8 181 (36.1) 11-Sep 263 (52.5) 12+ 501 A, B1, and B2 47 (9.4) C1 99 0 (18.0) C2 186 (37.1) D and E 1178 (35.5) Monthly per capita family income, BRI, Median (IQR) 421 1156 (880-1880) Reproductive history Previous pregnancy 501 Yes 320 (63.9) 10 No 181 (36.1) Children with malformations from previous pregnancies 501 Yes 12 (2.4) No 220 (63.9) No 181 (36.1) Children with malformations from previous pregnancies 501 Yes 12 (2.4) No 2278 (55.5) NA ^A 211 (42.1) 240 (93.0) <	Sociodemographic factors		
Race/ethnicity 500 "Preto" (ic, black) 50 (07.0) "Branco" (ic, white) 0107 (21.4) Other 8 (1.6) Years of education 501 0-8 181 (36.1) 11-Sep 263 (52.5) 12+ 0501 Social Class* 0 A. B1, and B2 47 (9.4) C1 90 (18.0) C2 188 (63.71) D and E 178 (35.5) Monthly per capita family income, BRL, Median (IQR) 421 Previous pregnancy 501 Yes 220 (63.9) No 181 (36.1) Children with malformations from previous pregnancies 501 Yes 12 (2.4) No 2278 (55.5) NA ^d 211 (42.1) Current pregnancy characteristics 12 Yes 38 (7.6) No 443 (92.4) Hicki drug use 501 Yes 7 (1.4) No 2494 (98.6) Delery mode	Age, years, Median (IQR)	501	26.2 (21-31)
"Preto" (ie, black) 50 (10.0) "Branco" (ie, mixed race) 335 (67.0) "Branco" (ie, white) 107 (21.4) Other 8 (16.6) Years of education 501 0-8 181 (36.1) 11-Sep 263 (52.5) 12+ 57 (11.4) Social Class" 501 C1 90 (18.0) C2 186 (37.1) Dand E 178 (35.5) Monthly per capita family income, BRL, Median (IQR) 421 Previous pregnancy 501 Yes 320 (63.9) No 181 (36.1) Children with malformations from previous pregnancies 501 Yes 122 (2.4) No 278 (55.5) N/A ^c 211 (2.1) Current pregnancy characteristics 38 (7.6) Smoking 501 Yes 38 (7.6) No 443 (92.4) No 443 (92.4) Yes 38 (7.6) N/A ^c 212 (2.4) No	Race/ethnicity	500	
"Pardo" (ie, mixed race) 335 (67.0) "Branco" (ie, white) 107 (21.4) Other 8 (1.6) Years of education 501 0-8 181 (36.1) 11-Sep 263 (52.5) 12+ 57 (11.4) Social Class ^b 501 A, B1, and B2 47 (9.4) C1 990 (18.0) C2 186 (37.1) D and E 178 (35.5) Monthly per capita family income, BRL, Median (IQR) 421 1156 (880-1880) Reproductive history 7 7 Previous pregnancy 501 7 Yes 320 (63.9) 181 (36.1) Children with malformations from previous pregnancies 501 7 Yes 12 (24) 10 (25.5) N/A ^c 211 (42.1) 12 (24) No 12 (24) 10 (26.55) N/A ^c 211 (42.1) 12 (24) No 278 (55.5) 12 (14.2) Urrent pregnancy characteristics 7 11 (42.1) So	"Preto" (ie, black)		50 (10.0)
"Branco" (ie, white) 107 (21.4) Other 8 (16) Years of education 501 0-8 181 (36.1) 11-Sep 263 (52.5) 12+ 57 (11.4) Social Class ^b 501 A, B1, and B2 47 (9.4) C1 90 (18.0) C2 186 (37.1) D and E 178 (35.5) Monthly per capita family income, BRL, Median (IQR) 421 1156 (880-1880) Reproductive history 2 181 (36.1) Yes 320 (63.9) 320 (63.9) No 181 (36.1) 181 (36.1) Children with malformations from previous pregnancies 501 278 (55.5) MA ^c 211 (22.4) No 228 (55.6) N/A ^c 211 (22.4) No 228 (55.6) Morking 501 278 (55.6) Yes 38 (7.6) 38 (7.6) No 463 (92.4) 116 (33 (92.4) Unicit drug use 501 227 (46.0) Yes 77 (1.4)	"Pardo" (ie, mixed race)		335 (67.0)
Other 8 (1.6) Yeas of education 501 0-8 181 (36.1) 11-Sep 263 (52.5) 12+ 57 (11.4) Social Class ^b 501 A, B1, and B2 47 (9.4) C1 90 (18.0) C2 186 (37.1) D and E 178 (35.5) Monthly per capita family income, BRL, Median (IQR) 421 Yes 320 (63.9) No 181 (36.1) Children with malformations from previous pregnancies 501 Yes 212 (24) No 278 (55.5) NA ^c 211 (42.1) Current pregnancy characteristics 211 (42.1) Smoking 501 Yes 38 (7.6) No 4463 (92.4) Itic targ use 501 Yes 7 (1.4) No 494 (98.6) Delivery mode 494 Cesarean 227 (46.0) Vaginal 266 (33.8) Forceps 10.02	"Branco" (ie, white)		107 (21.4)
Years of education 501 0-8 181 (36.1) 11-Sep 263 (52.5) 12+ 57 (11.4) Social Class ^b 501 A, B1, and B2 47 (9.4) C1 90 (18.0) C2 186 (37.1) D and E 178 (35.5) Monthly per capita family income, BRL, Median (IQR) 421 Previous pregnancy 501 Yes 320 (63.9) No 181 (36.1) Children with malformations from previous pregnancies 501 Yes 12 (2.4) No 227 (55.5) N/A ^c 211 (42.1) Current pregnancy characteristics 211 (42.1) Smoking 501 Yes 38 (7.6) No 463 (92.4) Yes 7 (1.4) No 494 (98.6) Delivery mode 494 Cesarean 227 (46.0) Yes 10.2 ZIKV diagnosis in pregnancy 501 ZIKV-positive 22	Other		8 (1.6)
0-8 181 (36.1) 11-Sep 263 (52.5) 12+ 57 (11.4) Social Class ^b 501 A, B1, and B2 47 (9.4) C1 90 (18.0) C2 186 (37.1) D and E 178 (35.5) Monthly per capita family income, BRL, Median (IQR) 421 1156 (880-1880) Reproductive history 2 186 (37.1) Previous pregnancy 501 2 Yes 320 (63.9) 320 (63.9) No 181 (36.1) 181 (36.1) Children with malformations from previous pregnancies 501 278 (55.5) N/A ^c 211 (42.1) 278 (55.5) N/A ^c 211 (42.1) 278 (55.5) V/A ^c 211 (42.1) 38 (7.6) Quirrent pregnancy characteristics 38 (7.6) 38 (7.6) No 4463 (92.4) 463 (92.4) Users 501 227 (46.0) Yes 7 (1.4) No 494 (98.6) Delivery mode 494 226 (53.8) </td <td>Years of education</td> <td>501</td> <td></td>	Years of education	501	
11-Sep 263 (52.5) 12+ 57 (11.4) Social Class ^b 501 A, B1, and B2 47 (9.4) Cl 90 (18.0) C2 186 (37.1) D and E 178 (35.5) Monthly per capita family income, BRL, Median (IQR) 421 Previous pregnancy 1 Yes 320 (63.9) No 181 (36.1) Children with malformations from previous pregnancies 501 Yes 12 (2.4) No 278 (55.5) N/A ^c 2111 (42.1) Current pregnancy characteristics 2 Yes 38 (7.6) No 463 (92.4) Ilicit drug use 501 Yes 7 (1.4) No 494 (98.6) Delivery mode 494 Cesarean 227 (46.0) Vaginal 266 (53.8) Forceps 1 (0.2) ZIKV diagnosis in pregnancy 501 ZIKV-positive 277 (55.3) Robust evidence, serol	0-8		181 (36.1)
12+ 57(11.4) Scial Class ^b 501 A, B1, and B2 47(9.4) C1 90(18.0) C2 188(37.1) D and E 178(35.5) Monthly per capita family income, BRL, Median (IQR) 421 1156 (880-1880) Reproductive history Previous pregnancy 501 Yes 320 (63.9) No 181 (36.1) Yes 12 (2.4) No 278 (55.5) N/A ^c 2211 (42.1) Current pregnancy characteristics 278 (55.5) Smoking 501 Yes 38 (7.6) 38 (7.6) No 463 (92.4) Yes 7 (1.4) 38 (7.6) Yes 7 (1.4) Yes 7 (1.4) Yes 7 (1.4)	11-Sep		263 (52.5)
Social Class ^b 501 A, B1, and B2 47 (9.4) C1 00 (18.0) C2 186 (37.1) D and E 178 (35.5) Monthly per capita family income, BRL, Median (IQR) 421 1156 (880-1880) Reproductive history 2 320 (63.9) Previous pregnancy 501 181 (36.1) Children with malformations from previous pregnancies 501 278 (55.5) NA 12 (2.4) No 278 (55.5) NA ^c 211 (42.1) 278 (55.5) 211 (42.1) Current pregnancy characteristics 2 278 (55.5) NA ^c 211 (42.1) Current pregnancy characteristics 38 (7.6) 38 (7.6) 38 (7.6) 38 (7.6) No 463 (92.4) 1116 (14.9) 39.0 (63.8) 227 (46.0) 227 (46.0) Yes 71 (14.2) No 494 227 (46.0) 227 (46.0) 227 (46.0) 227 (46.0) 227 (46.0) 227 (46.0) 227 (46.0) 227 (46.0) 233 235 (23.8) 235 (23.8) 235 (23.8)	12+		57 (11.4)
A, B1, and B2 47 (9.4) C1 99 (18.0) C2 186 (37.1) D and E 178 (35.5) Monthly per capita family income, BRL, Median (IQR) 421 1156 (880-1880) Reproductive history 2 Previous pregnancy 501 181 (36.1) Children with malformations from previous pregnancies 501 12 (2.4) No 121 (2.4) No 278 (55.5) N/A ^c 211 (42.1) 278 (55.5) 14 (42.1) Current pregnancy characteristics 211 (42.1) 278 (55.5) 14 (42.1) Current pregnancy characteristics 211 (42.1) 278 (55.5) 14 (42.1) Current pregnancy characteristics 211 (42.1) 278 (55.5) 14 (42.1) Monthing 43 (92.4) 116 (43 (92.4) 116 (42.1) 116 (42.1) Ves 501 277 (55.3) 14 (42.1) 116 (42.1) 116 (42.1) Unitic drug use 501 277 (46.0) 14 (98.6) 14 (98.6) 16 (98.2) 16 (98.2) 10 (02.2) 116 (49.6) 16 (42.	Social Class ^b	501	
C1 90 (18.0) C2 186 (37.1) D and E 178 (35.5) Monthly per capita family income, BRL, Median (IQR) 421 1156 (880-1880) Reproductive history 90 90 1156 (880-1880) Previous pregnancy 501 90 1156 (880-1880) Yes 320 (63.9) No 181 (36.1) Children with malformations from previous pregnancies 501 90 12 (2.4) No 278 (55.5) N/4 ^c 211 (42.1) Yes 278 (55.5) 12 (2.4) No 278 (55.5) N/A ^c 211 (42.1) 12 (2.4) No 278 (55.5) Ves 320 (63.9) 38 (7.6) 38 (7.6) No 463 (92.4) 11 (42.1) 11 (42.1) Current pregnancy characteristics 38 (7.6) 10 (2.2) Yes 501 12 (2.4) No 443 (98.6) Ves 17 (1.4) No 494 (98.6) 226 (53.8) Porceps 10 (0.2) 227 (46.0) 226 (53.8)	A, B1, and B2		47 (9.4)
C2 186 (37.1) D and E 178 (35.5) Monthly per capita family income, BRL, Median (IQR) 421 1156 (880-1880) Reproductive history 2 1156 (880-1880) Previous pregnancy 501 9 Yes 320 (63.9) 181 (36.1) Children with malformations from previous pregnancies 501 12 (2.4) No 278 (55.5) N/A ^c 211 (42.1) Current pregnancy characteristics 2 38 (7.6) No 463 (92.4) 38 (7.6) Yes 38 (7.6) 227 (46.0) Yes 7 (1.4) No 494 (98.6) Delivery mode 494 227 (46.0) Vaginal 2266 (53.8) Forceps Forceps 1 (0.2) 21K V diagnosis in pregnancy ZIKV positive 277 (55.3) 33 Robust evidence, serology 53 33 Moderate evidence, serology 53 33 Limited vidence, serology 53 33 Limited vidence, serology 53	Cl		90 (18.0)
D and E 178 (35.5) Monthly per capita family income, BRL, Median (IQR) 421 1156 (880-1880) Reproductive history 2 320 (63.9) Previous pregnancy 501 320 (63.9) No 181 (36.1) 181 (36.1) Children with malformations from previous pregnancies 501 12 (2.4) No 278 (55.5) N/A ^c 211 (42.1) Current pregnancy characteristics 2 2 Smoking 501 38 (7.6) Yes 38 (7.6) 38 (7.6) No 4463 (92.4) 11 Yes 501 2 Yes 7 (1.4) No 4463 (92.4) No 4463 (92.4) 10 227 (46.0) Yes 501 2 227 (46.0) Yes 501 227 (46.0) 226 (53.8) Forceps 1 (0.2) 227 (46.0) 226 (53.8) Forceps 1 (0.2) 227 (46.0) 226 (53.8) Forceps 1 (0.2) 227 (55.3) 33	C2		186 (37.1)
Monthly per capita family income, BRL, Median (IQR) 421 1156 (880–1880) Reproductive history 501	D and E		178 (35.5)
Reproductive history 501 Yes 320 (63.9) No 181 (36.1) Children with malformations from previous pregnancies 501 Yes 12 (2.4) No 278 (55.5) N/A ^c 211 (42.1) Current pregnancy characteristics 211 (42.1) Yes 38 (7.6) No 463 (92.4) Illicit drug use 501 Yes 7 (1.4) No 4463 (92.4) Illicit drug use 501 Yes 7 (1.4) No 4494 (98.6) Delivery mode 494 Cesarean 227 (46.0) Vaginal 266 (53.8) Forceps 1 (0.2) ZIKV positive 277 (55.3) Robust evidence, PCR 108 Robust evidence, serology 53 Moderate evidence, serology 53 Limited evidence, serology 53 Limited evidence, serology 53 Limited evidence, serology 53	Monthly per capita family income, BRL, Median (IQR)	421	1156 (880–1880)
Previous pregnancy 501 Yes 320 (63.9) No 181 (36.1) Children with malformations from previous pregnancies 501 Yes 12 (2.4) No 278 (55.5) N/A ^c 211 (42.1) Current pregnancy characteristics 211 (42.1) Current pregnancy characteristics 38 (7.6) Yes 38 (7.6) No 463 (92.4) Illicit drug use 501 Yes 7 (1.4) No 494 (98.6) Delivery mode 494 Cesarean 227 (46.0) Vaginal 2266 (53.8) Forceps 1 (0.2) ZIKV diagnosis in pregnancy 501 ZIKV positive 277 (55.3) Robust evidence, PCR 108 Robust evidence, serology 53 Jumited evidence, serology 53 Limited evidence, serology 53 Jumited evidence, serology 53 Jumited evidence, serology 53 Jumited evidence, serol	Reproductive history		
Yes 320 (63.9) No 181 (36.1) Children with malformations from previous pregnancies 501 Yes 12 (2.4) No 278 (55.5) N/A ^c 211 (42.1) Current pregnancy characteristics 211 (42.1) Smoking 501 Yes 38 (7.6) No 463 (92.4) Illicit drug use 501 Yes 7 (1.4) No 494 (98.6) Delivery mode 494 Cesarean 227 (46.0) Vaginal 2266 (53.8) Forceps 1 (0.2) ZIKV diagnosis in pregnancy 501 ZIKV positive 277 (55.3) Robust evidence, PCR 108 Robust evidence, serology 533 Junipecified flavivirus-positive 733 Inconclusive 78 (15.6) Inconclusive 19 (38) ZIKV-negative 127 (25.3)	Previous pregnancy	501	
No 181 (36.1) Children with malformations from previous pregnancies 501 Yes 12 (2.4) No 278 (55.5) N/A ^c 211 (42.1) Current pregnancy characteristics 211 Smoking 501 Yes 38 (7.6) No 463 (92.4) Illicit drug use 501 Yes 7 (1.4) No 494 (98.6) Delivery mode 494 Cesarean 227 (46.0) Vaginal 266 (53.8) Forceps 11 (0.2) ZIKV diagnosis in pregnancy 501 ZIKV positive 277 (55.3) Robust evidence, PCR 108 Robust evidence, serology 533 Moderate evidence, serology 533 Moderate evidence, serology 533 Limited evidence, serology 533 Limited evidence, serology 533 Junspecified flavivirus-positive 78 (15.6) Inconclusive 19 (3.8) ZIKV-negative	Yes		320 (63.9)
Children with malformations from previous pregnancies501Yes12 (2.4)No278 (55.5)N/A ^c 211 (42.1)Current pregnancy characteristics201Smoking501Yes38 (7.6)No463 (92.4)Illicit drug use501Yes7 (1.4)No494 (98.6)Delivery mode494Cesarean227 (46.0)Vaginal266 (53.8)Forceps11 (0.2)ZIKV diagnosis in pregnancy501ZIKV positive277 (55.3)Robust evidence, PCR108Robust evidence, serology533Moderate evidence, serology533Unspecified flavivirus-positive78 (15.6)Inconclusive19 (3.8)ZIKV-negative19 (3.8)ZIKV-negative127 (25.3)	No		181 (36.1)
Yes 12 (2.4) No 278 (55.5) N/A ^c 211 (42.1) Current pregnancy characteristics 211 Smoking 501 Yes 38 (7.6) No 463 (92.4) Illicit drug use 501 Yes 7 (1.4) No 494 (98.6) Delivery mode 494 Cesarean 227 (46.0) Vaginal 266 (53.8) Forceps 1 (0.2) ZIKV diagnosis in pregnancy 501 ZIKV-positive 277 (55.3) Robust evidence, PCR 108 Robust evidence, serology 53 Limited evidence, serology 53 Limited evidence, serology 53 ZIKV-negative 78 (15.6) Inconclusive 19 (3.8) ZIKV-negative 127 (25.3)	Children with malformations from previous pregnancies	501	
No 278 (55.5) N/A ^c 211 (42.1) Current pregnancy characteristics 211 Smoking 501 Yes 38 (7.6) No 463 (92.4) Illicit drug use 501 Yes 7 (1.4) No 494 (98.6) Delivery mode 494 Cesarean 227 (46.0) Vaginal 266 (53.8) Forceps 1 (0.2) ZIKV diagnosis in pregnancy 501 ZIKV-positive 277 (55.3) Robust evidence, PCR 108 Robust evidence, serology 533 Moderate evidence, serology 533 Limited evidence, serology 533 ZIKV-negative 78 (15.6) Inconclusive 19 (3.8) ZIKV-negative 127 (25.3)	Yes		12 (2.4)
N/A ^c 211 (42.1) Current pregnancy characteristics 501 Smoking 501 Yes 38 (7.6) No 463 (92.4) Illicit drug use 501 Yes 7 (1.4) No 494 (98.6) Delivery mode 494 Cesarean 227 (46.0) Vaginal 266 (53.8) Forceps 1 (0.2) ZIKV diagnosis in pregnancy 501 ZIKV-positive 277 (55.3) Robust evidence, PCR 108 Robust evidence, serology 533 Unspecified flavivirus-positive 78 (15.6) Inconclusive 78 (15.6) ZIKV-negative 127 (25.3)	No		278 (55.5)
Current pregnancy characteristics 501 Smoking 501 Yes 38 (7.6) No 463 (92.4) Illicit drug use 501 Yes 501 Yes 7 (1.4) No 494 (98.6) Delivery mode 494 Cesarean 227 (46.0) Vaginal 266 (53.8) Forceps 1 (0.2) ZIKV diagnosis in pregnancy 501 ZIKV-positive 2177 (55.3) Robust evidence, PCR 108 Robust evidence, serology 533 Unspecified flavivirus-positive 78 (15.6) Inconclusive 19 (3.8) ZIKV-negative 127 (25.3)	N/A ^c		211 (42.1)
Smoking 501 Yes 38 (7.6) No 463 (92.4) Illicit drug use 501 Yes 501 Yes 7 (1.4) No 494 (98.6) Delivery mode 494 Cesarean 227 (46.0) Vaginal 266 (53.8) Forceps 1 (0.2) ZIKV diagnosis in pregnancy 501 ZIKV-positive 277 (55.3) Robust evidence, PCR 108 Robust evidence, serology 53 Limited evidence, serology 53 Unspecified flavivirus-positive 78 (15.6) Inconclusive 19 (3.8) ZIKV-negative 127 (25.3)	Current pregnancy characteristics		
Yes 38 (7.6) No 463 (92.4) Illicit drug use 501 Yes 7 (1.4) No 494 (98.6) Delivery mode 494 Cesarean 227 (46.0) Vaginal 266 (53.8) Forceps 1 (0.2) ZIKV diagnosis in pregnancy 501 ZIKV-positive 277 (55.3) Robust evidence, PCR 108 Robust evidence, serology 53 Limited evidence, serology 53 Unspecified flavivirus-positive 78 (15.6) Inconclusive 19 (3.8) ZIKV-negative 127 (25.3)	Smoking	501	
No 463 (92.4) Illicit drug use 501 Yes 7 (1.4) No 494 (98.6) Delivery mode 494 Cesarean 227 (46.0) Vaginal 266 (53.8) Forceps 1 (0.2) ZIKV diagnosis in pregnancy 501 ZIKV-positive 277 (55.3) Robust evidence, PCR 108 Robust evidence, serology 53 Moderate evidence, serology 53 Limited evidence, serology 53 Unspecified flavivirus-positive 78 (15.6) Inconclusive 19 (3.8) ZIKV-negative 127 (25.3)	Yes		38 (7.6)
Illicit drug use 501 Yes 7 (1.4) No 494 (98.6) Delivery mode 494 Cesarean 227 (46.0) Vaginal 266 (53.8) Forceps 1 (0.2) ZIKV diagnosis in pregnancy 501 ZIKV-positive 277 (55.3) Robust evidence, PCR 108 Robust evidence, serology 53 Limited evidence, serology 53 Unspecified flavivirus-positive 78 (15.6) Inconclusive 19 (3.8) ZIKV-negative 127 (25.3)	No		463 (92.4)
Yes 7 (1.4) No 494 (98.6) Delivery mode 494 Cesarean 227 (46.0) Vaginal 266 (53.8) Forceps 1 (0.2) ZIKV diagnosis in pregnancy 501 ZIKV-positive 277 (55.3) Robust evidence, PCR 108 Robust evidence, serology 53 Limited evidence, serology 53 Unspecified flavivirus-positive 78 (15.6) Inconclusive 19 (3.8) ZIKV-negative 127 (25.3)	Illicit drug use	501	
No 494 (98.6) Delivery mode 494 Cesarean 227 (46.0) Vaginal 266 (53.8) Forceps 1 (0.2) ZIKV diagnosis in pregnancy 501 ZIKV-positive 277 (55.3) Robust evidence, PCR 108 Robust evidence, serology 53 Moderate evidence, serology 53 Limited evidence, serology 53 Unspecified flavivirus-positive 78 (15.6) Inconclusive 19 (3.8) ZIKV-negative 127 (25.3)	Yes		7 (1.4)
Delivery mode494Cesarean227 (46.0)Vaginal266 (53.8)Forceps1 (0.2)ZIKV diagnosis in pregnancy501ZIKV-positive277 (55.3)Robust evidence, PCR108Robust evidence, serology53Moderate evidence, serology53Limited evidence, serology53Unspecified flavivirus-positive78 (15.6)Inconclusive19 (3.8)ZIKV-negative127 (25.3)	No		494 (98.6)
Cesarean227 (46.0)Vaginal266 (53.8)Forceps1 (0.2)ZIKV diagnosis in pregnancy501ZIKV-positive277 (55.3)Robust evidence, PCR108Robust evidence, serology53Moderate evidence, serology53Limited evidence, serology53Unspecified flavivirus-positive78 (15.6)Inconclusive19 (3.8)ZIKV-negative127 (25.3)	Delivery mode	494	
Vaginal266 (53.8)Forceps1 (0.2)ZIKV diagnosis in pregnancy501ZIKV-positive277 (55.3)Robust evidence, PCR108Robust evidence, serology53Moderate evidence, serology53Limited evidence, serology53Unspecified flavivirus-positive78 (15.6)Inconclusive19 (3.8)ZIKV-negative127 (25.3)	Cesarean		227 (46.0)
Forceps1 (0.2)ZIKV diagnosis in pregnancy501ZIKV-positive277 (55.3)Robust evidence, PCR108Robust evidence, serology53Moderate evidence, serology53Limited evidence, serology53Unspecified flavivirus-positive78 (15.6)Inconclusive19 (3.8)ZIKV-negative127 (25.3)	Vaginal		266 (53.8)
ZIKV diagnosis in pregnancy501ZIKV-positive277 (55.3)Robust evidence, PCR108Robust evidence, serology53Moderate evidence, serology53Limited evidence, serology53Unspecified flavivirus-positive78 (15.6)Inconclusive19 (3.8)ZIKV-negative127 (25.3)	Forceps		1 (0.2)
ZIKV-positive277 (55.3)Robust evidence, PCR108Robust evidence, serology53Moderate evidence, serology53Limited evidence, serology53Unspecified flavivirus-positive78 (15.6)Inconclusive19 (3.8)ZIKV-negative127 (25.3)	ZIKV diagnosis in pregnancy	501	
Robust evidence, PCR108Robust evidence, serology53Moderate evidence, serology53Limited evidence, serology53Unspecified flavivirus-positive78 (15.6)Inconclusive19 (3.8)ZIKV-negative127 (25.3)	ZIKV-positive		277 (55.3)
Robust evidence, serology53Moderate evidence, serology53Limited evidence, serology53Unspecified flavivirus-positive78 (15.6)Inconclusive19 (3.8)ZIKV-negative127 (25.3)	Robust evidence, PCR		108
Moderate evidence, serology53Limited evidence, serology53Unspecified flavivirus-positive78 (15.6)Inconclusive19 (3.8)ZIKV-negative127 (25.3)	Robust evidence, serology		53
Limited evidence, serology53Unspecified flavivirus-positive78 (15.6)Inconclusive19 (3.8)ZIKV-negative127 (25.3)	Moderate evidence, serology		53
Unspecified flavivirus-positive78 (15.6)Inconclusive19 (3.8)ZIKV-negative127 (25.3)	Limited evidence, serology		53
Inconclusive 19 (3.8) ZIKV-negative 127 (25.3)	Unspecified flavivirus-positive		78 (15.6)
ZIKV-negative 127 (25.3)	Inconclusive		19 (3.8)
	ZIKV-negative		127 (25.3)

Table 1. Characteristics of women with rash in the MERG Pregnancy Cohort in Pernambuco, Brazil (2015–2017).

^aNumbers varied because of missing values.

^bBased on the pattern of consumption of the Brazilian families, A being the higher and E being the lower. ^cNot applicable due to abortion, stillbirth, or no previous pregnancy.

https://doi.org/10.1371/journal.pntd.0009216.t001

Clinical Features	ZIKV-positive	Suspected ZIKV	exposure	ZIKV-negative (n = 127)	P value ^b Positive v. negative	P value ^c Positive +Suspected v. negative	
	Robust + Moderate + Limited (n = 277)	Unspecified flavivirus- positive (n = 78)	Inconclusive (n = 19)				
		N (%)		÷	-		
Rash	273 (98.6)	78 (100)	17 (84.2)	123 (96.8)	0.27	0.48	
Location where rash first appeared							
Head/neck	19 (6.9)	6 (7.7)	0 (0)	4 (3.1)	0.17	0.19	
Torso/back	46 (16.6)	12 (15.4)	1 (5.3)	27 (21.3)	0.26	0.16	
Abdomen	127 (45.8)	38 (48.7)	9 (47.4)	53 (41.7)	0.44	0.35	
Arms	60 (21.7)	18 (23.1)	9 (47.4)	29 (22.8)	0.79	0.92	
Legs	43 (15.5)	10 (12.8)	1 (5.3)	16 (12.6)	0.44	0.60	
Full body	18 (6.5)	7 (9.0)	0 (0)	7 (5.5)	0.70	0.64	
Duration of rash, days (Median (IQR))	4.5 (3-7)	4 (3-7)	5 (2-15)	5 (3–7)	0.29	0.23	
Fever	220 (80.6)	64 (82.0)	11 (68.7)	97 (78.9)	0.69	0.71	
Myalgia	197 (71.1)	47 (60.3)	10 (52.6)	82 (64.6)	0.19	0.49	
Arthralgia	236 (85.2)	65 (83.3)	12 (63.2)	96 (75.6)	0.02	0.04	
Joint swelling	196 (70.8)	51 (65.4)	9 (47.4)	76 (59.8)	0.03	0.08	
Retro-bulbar pain	78 (28.2)	18 (23.1)	6 (31.6)	39 (30.7)	0.60	0.46	
Secretions in the eyes when waking up	20 (7.2)	5 (6.4)	0 (0)	3 (2.4)	0.06	0.07	
Conjunctivitis	75 (27.1)	16 (20.5)	4 (21.0)	38 (29.9)	0.55	0.32	
Photophobia	71 (25.6)	15 (29.2)	5 (26.3)	43 (33.9)	0.09	0.04	
Headache	215 (77.6)	62 (79.5)	14 (73.7)	91 (71.6)	0.19	0.16	
Vomiting/nausea	117 (42.2)	34 (43.6)	7 (36.8)	52 (40.9)	0.81	0.80	
Rhinorrhea	37 (13.4)	10 (12.8)	2 (10.5)	10 (7.9)	0.11	0.11	
Pruritis	232 (83.8)	63 (80.8)	13 (68.4)	108 (85.0)	0.74	0.49	
Back pain	181 (65.3)	48 (61.5)	8 (42.1)	78 (61.4)	0.44	0.69	
Fatigue	219 (79.1)	62 (79.5)	13 (68.4)	100 (78.7)	0.94	0.97	
Pharyngitis	34 (12.3)	10 (12.8)	2 (10.5)	10 (7.9)	0.19	0.17	
Cough	29 (10.5)	7 (9.0)	2 (10.5)	10 (7.9)	0.41	0.45	
Abdominal pain	77 (27.8)	17 (21.8)	6 (31.6)	31 (24.4)	0.47	0.61	

Table 2. Clinical presentation of women with rash at the time of notification by ZIKV diagnostic status in the MERG Pregnancy Cohort in Pernambuco, Brazil (2015–2017)^a.

^aThe numbers varied because of the missing values.

^bComparison of the ZIKV-positive group versus the ZIKV-negative group: *P* values for categorical variables are from Chi-squared tests or, for variables with cells including \leq 5 observations, Fisher's exact tests; *P* values for continuous variables are from Mann-Whitney U tests.

^cComparison of the positive and suspected exposure group (i.e., ZIKV-positive, unspecified flavivirus positive, and inconclusive groups) versus the ZIKV-negative group: *P* values for categorical variables are from Chi-squared tests or, for variables with cells including \leq 5 observations, Fisher's exact tests; *P* values for continuous variables are from Mann-Whitney U tests.

https://doi.org/10.1371/journal.pntd.0009216.t002

Microcephaly and TF-US suggestive of CZS occurred independently of the trimester in which there was a probable Zika infection (Table 4).

A total of 14 children out of 277 had neurological alterations; 10 were born to women positive to ZIKV infection, 2 born to a woman who had unspecified flavivirus infection, one born to a woman with inconclusive results and one whose mother did not have laboratory evidence of ZIKV infection (Table 3). The later presented convulsions. Among those 10 born to women positive to ZIKV infection, 7 children had only neurological alterations; two children had

		Positive		Suspected ZIKV exposure			Negative	p-value (all positive	p-value (all		
	Cases/ sample ^c	Robust	Moderate	Limited	Robust + Moderate + Limited	Flavivírus	Inconclusive	Positive + Flavivírus + inconclusive		x negative)	suspected x negative)
Microcephaly	12/503	2/162 (1.2) 0.03- 4.9	1/28 (3.6) 0.4–23.7	4/88 (4.5) 1.7–11.7	7/278 (2.5) 1.2–5.2	3/78 (3.8) 1.2–11.5	1/20 (5.0) 0.6–32.3	11/376 (2.9) 1.6-5.2	1/127 (0.8) 0.1–5.2	0.453	0.301
Altered transfontanelle US ^a	18/319	5/84 (5.9) 2.4– 13.7	4/24 (16.7) 5.9–38.9	2/56 (4.3) 0.8–13.7	11/164 (6.7) 3·7-11·8	2/46 (4.3) 1.0–16.5	0/18 (0)	13/180 (7.2) 3.3-9.6	5/91 (5.5) 2.3–12.7	0.702	0.590
Neurologic Abnor	malities ^b										
At first evaluation	14/277	6/144 (4.2) 1.9–9.0	0/20 (0)	4/52 (7.7) 2.8–19.3	10/216 (4.6) 2.5-8.4	2/20 (10.0) 2.2–35.5	1/9 (11.1) 0.9–62.6	13/245 (5.3) 3.1-8.9	1/32 (3.1) 0.4–20.9	1.000	1.000
At any time	-	-	-	-	28/216 (13.0) 9.1–18.2	-	-	34/245 (13.9) 10.0–18.8	4/32 (12.5) 4.5-30.2	1.000	1.000
Ophthalmologic A	bnormalit	ies									
At first evaluation	20/243	6/143 (4.2) 1.9–9.1	1/20 (5.0) 0.6–32.3	5/24 (20.8) 8.3–43.2	12/187 (6.4) 3.7-11.0	2/18 (11.1) 2.8–38.9	1/9 (11.1) 0.09–62.6	15/214 (7.0) 4.2–11.3	5/29 (17.2) 0.07- 36.7	0,118	0,145
At any time	-	-	-	-	14/187 (7.5) 4.5–12.3	-	-	18/214 (8.4) 5.3–13.0	6/29 (20.7) 9.1-40.4	0.069	0.097
Abnormal auditory	v brainstem	response									
At first evaluation	11 ^d /253	7/133 (5.3) 2.5- 10.7	1/18 (5.6) 0.6–35.5	2/48 (4.2) 1.0–15.9	10/199 (5.0) 2.7–9.1	1/17 (5.9) 0.7–37.2	0/7 (0)	11/223 (4.9) 2.7-8.7	0/30 (0)	0.367	0.371
At any time	-	-	-	-	10/199 (5.0) 2.7–9.1	-	-	11/223 (4.9) 2.7-8.7	0/30 (0)	0.367	0.371
Dysphagia											
At first evaluation	4/256	2/132 (1.5) 0.4–5.9	0/18 (0)	2/50 (4.0) 1.0-15.3	4/200 (2.0) 0.7–5.2	0/17 (0)	0/9 (0)	4/226 (1.8) 0.7-4.7	0/30 (0)	1.000	1.000
At any time	-	-	-	-	6/200 (3.0) 1.3-6.6	-	-	6/226 (2.6) 1.2-5.8	0/30 (0)	0.856	0.939

Table 3. Absolute risk and 95% confidence interval of Zika related adverse outcomes in the MERG Pregnancy Cohort in Pernambuco, Brazil (2015-2017).

^aCalcification or ventriculomegaly.

^bAt least one of the following: inappropriate visual response, alteration of tonus or trophism, altered level of consciousness/behavior and convulsions.

^cThe denominator varies according to the missing values for each outcome.

^DNine children repeated the test and it was normal on the second evaluation.

https://doi.org/10.1371/journal.pntd.0009216.t003

neurologic alterations and microcephaly (one with alteration of the optic nerve and orthopaedic alterations). Alteration of the optic nerve was also observed in one child with no microcephaly.

The absolute risk of neurologic alterations among children born to women positive to ZIKV infection was 4.6% (CI-95%: 2.5–8.4%) (Table 3). Taking the alterations separately the risk of altered level of consciousness/behaviour, localized motor deficit, convulsion, inadequate visual response and sign of pyramidal release was less than 2%, while the risk of tonus alteration was slightly higher (3.5%, CI-95%: 1.6–7.6%).

Trimester of rash	Number of children ^b	Number of cases (%)	95%-CI	Relative risk (95%-CI)	p-value
Microcephaly	278	7			
First	38	1 (1.8)	0.2-10.0	Reference	-
Second	86	3 (2.2)	0.7-6.6	1.33 (0.14-36.1)	1.000
Third	154	7 (3.8)	1.8-7.8	1.73 (0.22–13.6)	1.000
Any of the abnormalities ^a	185	45			
First	32	8 (25.0)	12.5-43.7	Reference	-
Second	63	7 (11.1)	5.3-21.9	0.44 (0.18-1.12)	0.134
Third	90	21 (23.3)	15.6-33.4	0.93 (0.46-1.89)	0.814

Table 4. Gestational age of rash and risk of Zika related adverse outcomes in the Zika positive group of the MERG Pregnancy Cohort in Pernambuco, Brazil (2015–2017).

^aTotal number of children evaluated for at least one of the following: microcephaly, CNS imaging abnormalities, Neurologic Abnormalities, Ophthalmologic abnormalities.

^bThe denominator varies according to the missing values for each outcome.

https://doi.org/10.1371/journal.pntd.0009216.t004

A total of 15 out 214 children born to non-negative mothers had ophthalmologic alterations compatible with ZIKV infection, corresponding to an absolute risk of 7% (CI-95%: 4.2% to 11.3%) (Table 3). Taking each of the alterations separately, the risk ranged from 1% (strabismus, nystagmus) to 4.2% (optic nerve).

The absolute risk of dysphagia among children born to women positive to ZIKV infection was around 2% (Table 3). One of these children had microcephaly, neurologic, ophthalmologic alterations and imaging abnormalities (identified in the TF-US, computed tomography and magnetic resonance). Another had neurologic and ophthalmologic alterations. In the two others no other alterations were detected.

A total of 253 children undergone a hearing screening evaluation and 11 failed, none of the later were born to ZIKV negative mothers. However, nine of these children were retested and they all passed in the second evaluation.

There was no statistically significant difference in the risk of adverse outcomes between children born to mothers positive or with suspected Zika exposure and children born to negative mothers.

The percentage of children that had at least one of the abnormalities compatible with CZS was around 20% (Table 5), considering either children born to mothers who were positive, or non-negative (adding mothers with unspecified flavivirus or inconclusive results). The intersection between these different abnormalities did not exceed 1%.

Approximately two-thirds of the pregnant women notified with rash belonged to the lower social classes and only a small proportion had higher education (Table 1). There was no association between Zika related adverse outcomes and socioeconomic conditions (S2 Table).

Infection by other arbovirus or TORCH during pregnancy by ZIKV diagnostic status are presented in <u>S3 Table</u>. The frequency of TORCH was below 1%. Approximately 10% tested IgM positive for Dengue and 30% for Chikungunya.

Discussion

The absolute risk of occurrence of microcephaly in children born to non-negative mothers (positive and suspected) was approximately 3%. Most of the children with microcephaly who had a TF-US performed had no brain abnormality detected. The risk of calcifications and/or ventriculomegaly was 5.7%, of neurologic alterations was 4.6%, of ophthalmologic abnormalities was 6.3% and of dysphagia was around 2%. The percentage of children that had at least

Adverse outcomes	Number of children affected/ Total	Absolute risk (%)	95%-CI Absolute risk
Any of the outcomes			
Microcephaly and/or CNS imaging abnormalities and/or Neurologic Abnormalities and/or Ophthalmologic Abnormalities			
At first evaluation			
Positive	21/99	21.2	14.1-30.5
Positive + Flavivirus + inconclusive	26/119	21.8	15.2-30.3
At any time			
Positive	28/99	28.3	20.2-38.1
Positive + Flavivirus + inconclusive	37/119	31.1	23.3-40.1
Combined outcomes			
Microcephaly and CNS imaging abnormalities and Neurologic Abnormalities			
At first evaluation			
Positive ^a	1/121	0.8	0.1-5.8
Positive + Flavivirus + inconclusive	1/141	0.7	0.1-5.0
At any time			
Positive ^a	2/121	1.6	0.4-6.5
Positive + Flavivirus + inconclusive	2/141	1.4	0.3-5.6
Microcephaly and Neurologic Abnormalities and Ophthalmologic Abnormalities			
At first evaluation			
Positive	0/180	-	-
Positive + Flavivirus ^b + inconclusive	1/207	0.5	0.1-3.4
At any time			
Positive	0/180	-	-
Positive + Flavivirus ^b + inconclusive	1/207	0.5	0.1-3.4
CNS imaging abnormalities and Neurologic Abnormalities			
At first evaluation			
Positive ^c	1/120	0.8	0.1-5.8
Positive + Flavivirus + inconclusive	1/140	0.7	0.1-5.0
At any time			
Positive ^c	1/120	0.8	0.1-5.8
Positive + Flavivirus + inconclusive	1/140	0.7	0.1-5.0
Neurologic Abnormalities and Ophthalmologic Abnormalities			
At first evaluation			
Positive	1/180	0.5	0.1-3.9
Positive + Flavivirus + inconclusive	3/207	1.4	0.5-4.4
At any time			
Positive	3/180	1.7	0.5-5.1
Positive + Flavivirus + inconclusive	5/207	2.4	1.0-5,7

Table 5. Absolute risk of Zika related adverse outcomes in the MERG Pregnancy Cohort in Pernambuco, Brazil (2015–2017).

 ${}^{*} Total \ number \ of \ children \ evaluated \ for \ microcephaly, CNS \ imaging \ abnormalities, \ Neurologic \ Abnormalities, \ Ophthalmologic \ Abnormalities.$

^bFlavivirus.

^cRobust

https://doi.org/10.1371/journal.pntd.0009216.t005

one of the abnormalities compatible with CZS was around 20%. However, the intersection between these different abnormalities (microcephaly, imaging, neurologic and ophthalmologic) did not exceed 1%.

^aLimited.

The frequency of microcephaly is consistent with the results of other studies [6–11]. The frequency of at least one type of alteration compatible with prenatal exposure to ZIKV infection in our study lies between the 6% and 46% found by Hoen et al. [7] and Brasil et al. [6], respectively. However the comparability between studies is limited by the different clinical and laboratory criteria adopted to define exposure [6,11,27], the limitation of the diagnostic tests to detect ZIKV infection [18] and by the lack of standardisation to define and assess the outcomes [6–13]. Investigation of children in the different studies was similar in terms of microcephaly and imaging, but differed with other outcomes.

Our results suggest that there is not a single presentation of microcephaly, occurring with or without imaging abnormalities and with the presence, or not, of neurological, and ophthal-mologic alterations. In Brasil's cohort the four children with microcephaly had imaging abnormalities, half of them had disproportionate microcephaly [6]. However microcephaly without imaging abnormalities was also described in the French Territories [7], in the U.S. territories and freely associated states [10] and in the continental US and Hawaii [11]. The finding of microcephaly in a child born to a mother with no laboratory evidence of ZIKV infection may be due to false negative result, considering the limitations of the diagnostic tests available [18], and the fact that it occurred during the microcephaly epidemic by Zika [1]. However, another congenital infection cannot be excluded as this mother had rash during pregnancy and was not tested for TORCH. Microcephaly has several causes and is a rare event [28]. The annual mean of occurrence of cases in the state of Pernambuco in the period before the epidemic was 9 [1,29]. We did not investigate genetic and metabolic disorders and they cannot be discarded but are not likely to have occurred.

In our study the imaging abnormalities potentially associated with ZIKV infection were mainly calcifications and ventriculomegaly. A wider variation of abnormalities, particularly in children with microcephaly, has been registered [30] but calcifications and ventriculomegaly were among the most frequent. Intracranial calcifications may occur in other congenital infections like herpes simplex virus, cytomegalovirus and toxoplasmosis and excluding other aetiologies based on specific mother testing is not without problem.

The risk of neurologic alterations among children born to women with laboratory evidence of ZIKV infection during pregnancy was of 5% (CI-95%: 2.7–9.1%) occurring either isolated or in the presence of other type of alterations. The frequency was much lower than that of 26% described by Brasil et al. [6]. One possible explanation for this difference would be that in the study of Brasil et al. [6] children were evaluated at birth and, in our study, they were evaluated later; it is possible that some of these neurologic abnormalities described in the former study were transient.

We found a risk of ophthalmologic alterations of 6.3% (CI-95%: 3.1%-9.5%), very similar to the 6% of the cohort of Rio de Janeiro [6], the only other cohort which provides this information. The ocular abnormalities we found had been described previously, notably in children with microcephaly [31] but also in children without microcephaly [32].

The risk of dysphagia among children born to women with laboratory evidence of ZIKV infection during pregnancy was around 2%, found in children with or without other alterations. Hearing loss and dysphagia associated with congenital virus infection has been previously described, but only in children with microcephaly [33,34].

When we compared the frequency of abnormalities at birth or first evaluation with those that were observed at any time during the follow-up, we found that there was an increase only for the neurological abnormalities. It is not surprising as ophthalmologic abnormalities and dysphagia were already present at the age when the children were evaluated for the first time (median age of 4.65 months (IQR: 2.9–13.4)). However, some neurological abnormalities can

be more easily identified or may appear as children get older; for example, epilepsy, motor abnormalities and maturational milestones may not be detectable until later ages.

Overall, we found no statistically significant differences in the risks of adverse outcomes in children born to mothers with non-negative or negative ZIKV test results. However, these results need to be interpreted with caution as the latter group may not be truly negative for ZIKV infections. As this is a cohort of PW notified with rash, all participants had symptoms compatible with ZIKV infection during an active epidemic. In fact, in addition to rash, in the negative group, almost 80% of the mothers reported fever, arthralgia or fatigue and more than 60% reported myalgia, among other symptoms. Limitations of the accuracy of the diagnostic tests for ZIKV and the narrow time window in which the molecular tests remain positive [18] also increase the uncertainty in excluding this diagnosis [18]. Another point to be considered is that, due to financial and operational constrains, we prioritized the evaluation of children born to mothers who had laboratory evidence of ZIKV infection during pregnancy. Therefore, the number of children born to mothers who tested negative is small, especially in relation to neurologic and ophthalmologic assessments, and therefore the power to detect differences in the risk of abnormalities between children born to mothers of this group and those born to non-negative mothers is low. In summary, false negative results and lack of power may underly the lack of association between adverse outcomes and positive ZIKV test results.

The definition of CZS is still being built and the clinical features of the less severe cases are not well known. Our results suggest that abnormalities occur more frequently isolated than in combination. The frequency of any adverse outcome could be as high as one in five children whose mother was infected during pregnancy but the frequency of the concomitant presentation of microcephaly, imaging, neurologic and ophthalmologic abnormalities was very low. Even for the children with microcephaly, there was a degree of variation, including children with no other abnormality detected. The reason for differences in the presentation needs to be further explored. It would be expected more severe cases in children born to mothers infected in the first trimester of pregnancy as observed in some studies [7,8,11] and in other congenital diseases [35,36]. However, abnormalities were also found in children born to mothers who were infected later in pregnancy. Another point to learn the full spectrum is that some neuro-developmental abnormalities, which may be an important component of this Syndrome, will only be better evaluated later in these children.

As pregnant women were notified because of presence of rash this symptom was reported in the interview by 98% of them. The characteristics of the rash did not differ between those who had or not laboratory evidence of ZIKV infection and, in general, it was not possible to distinguish Zika infection just by clinical characteristics. The proportion of laboratory evidence of DENV and CHIKV infection among those who had no laboratory evidence of Zika was small.

Five infant deaths occurred after delivery, three of which, including two children with Down Syndrome, in children with severe cardiac malformation which are not likely to be related with ZIKV infection. For the other foetal or infant deaths it is not possible to evaluate whether ZIKV infection was a contributing factor.

Overall, the cohort had a low socioeconomic status, with approximately two-thirds of participants belonging to the lower social classes and only a small proportion of the mothers having higher education. An ecological study carried out by MERG showed that there was a strong association between a higher prevalence of microcephaly and poor living conditions in the Metropolitan Region of Recife [37]. In fact, this area is characterized by high social inequality, and the pattern of distribution of several infectious diseases reflects the uneven social conditions [38,39]. The frequency of TORCH (Toxoplasmosis, Rubella, Cytomegalovirus and Parvovirus) infections among the pregnant women was low, but at least one-third of those tested for Chikungunya were positive. We note that Zika and Chikungunya viruses were co-circulating in this area in the same period and the neurological manifestations of the dual infection in the adult population has been recently described [40]. We cannot rule out that CHIKV may be playing a role in some of the outcomes we found.

This study has some strengths and limitations. The partnership with health authorities by using the notification system in the recruitment of pregnant women with rash in the context of the public health emergency reduced the risk of selection bias. There was a higher health-seeking behaviour increasing the sensitivity in detecting women with rash. The use of different laboratory technics and sequenced tests allowed to compare hierarchical groups of evidence of ZIKV infection in a sensitive analysis. The use of longitudinal molecular and serologic testing enhanced diagnostic sensitivity for defining cases of ZIKV infection among pregnant women presenting with rash and allowed us to detect a larger number of infected women. Another strength was the standardized and deep investigation of infant outcomes, performed by experienced specialists.

The children were not evaluated at birth and the time elapsed until the first evaluation varied. Early transient symptoms may not have been detected, but the adverse outcomes which were diagnosed are likely to be persistent. Furthermore, as impairments may manifest at a later age, some abnormalities were only detected because of the delayed evaluation. Later manifestations point to the need to follow these children for a longer period.

Asymptomatic pregnant women were not included and it is possible that risk estimates for children born to asymptomatic pregnant women could differ but there is no evidence to date that it occurs [8,41].

Misclassification in relation to exposure may have happened due to the limitations of the laboratory tests, the short duration of viremia and the overlapping of the clinical features with other arbovirus infections and TORCHs. However, there was no marked difference in the frequency of adverse outcomes between the groups with different levels of evidence. Errors of classification may also have occurred in the unexposed group because they were symptomatic and therefore could have another infection.

Not all children were evaluated for all outcomes. However, as the children were not selected to the comprehensive evaluation for any particular reason, it may have decreased the sample size but it is not likely to have introduced any selection bias.

We may not exclude that there may have been some abnormalities noted in the infants born to mothers with laboratory evidence which have an aetiology unrelated to ZIKV. Other studies with an adequate control group of unexposed women would allow the estimation of the attributable risk.

Conclusions and recommendations

The frequency of microcephaly in our results is consistent with previous studies. In addition, as the children were evaluated at a later age and repeatedly by different groups of specialists our findings provide new insight into the absolute risk of more specific abnormalities, especially those more likely to be persistent, and suggest that isolated manifestations occur more frequently than in combination. Due to the rare nature of some outcomes and the possibility of later manifestations of some components of the syndrome, large scale individual participant data meta-analysis to explore subgroups and the long term evaluation of children are imperative to identify the full spectrum and to plan actions to reduce the damages of the Congenital Zika Syndrome. For the children with microcephaly, we recommend a comprehensive clinical

examination by the paediatrician to identify other adverse outcomes/complications and the early initiation of neuropsychomotor stimulation. For the other exposed but symptomatic or asymptomatic children, we recommend screening for neurologic and ophthalmologic abnormalities and developmental delay.

Supporting information

S1 Table. Characteristics of women with rash by ZIKV diagnostic status in the MERG Pregnancy Cohort in Pernambuco, Brazil (2015–2017). (DOCX)

S2 Table. Socioeconomic conditions and risk of Zika related adverse outcomes in the in the Zika positive group of the MERG Pregnancy Cohort in Pernambuco, Brazil (2015–2017). (DOCX)

S3 Table. Infection by other arbovirus or TORCH during pregnancy by ZIKV diagnostic status in the MERG Pregnancy Cohort in Pernambuco, Brazil (2015–2017). (DOCX)

Author Contributions

- **Conceptualization:** Ricardo Arraes de Alencar Ximenes, Demócrito de Barros Miranda-Filho, Ulisses Ramos Montarroyos, Celina Maria Turchi Martelli, Thalia Velho Barreto de Araújo, Elizabeth Brickley, Maria de Fátima Pessoa Militão de Albuquerque, Wayner Vieira Souza, Liana O. Ventura, Camila V. Ventura, Adriana L. Gois, Mariana C. Leal, Danielle Maria da Silva Oliveira, Sophie Helena Eickmann, Maria Durce C. G. Carvalho, Paula F. S. da Silva, Maria Angela Wanderley Rocha, Regina Coeli Ferreira Ramos, Sinval Pinto Brandão-Filho, Marli Tenorio Cordeiro, Luciana Caroline Albuquerque Bezerra, George Dimech, Sandra Valongueiro, Pedro Pires, Priscila Mayrelle da Silva Castanha, Rafael Dhalia, Ernesto Torres Azevedo Marques-Júnior, Laura C. Rodrigues.
- **Data curation:** Ricardo Arraes de Alencar Ximenes, Demócrito de Barros Miranda-Filho, Ulisses Ramos Montarroyos, Celina Maria Turchi Martelli, Thalia Velho Barreto de Araújo, Maria de Fátima Pessoa Militão de Albuquerque, Wayner Vieira Souza, Laura C. Rodrigues.
- Formal analysis: Ricardo Arraes de Alencar Ximenes, Demócrito de Barros Miranda-Filho, Ulisses Ramos Montarroyos, Celina Maria Turchi Martelli, Thalia Velho Barreto de Araújo, Elizabeth Brickley, Laura C. Rodrigues.
- **Funding acquisition:** Ricardo Arraes de Alencar Ximenes, Demócrito de Barros Miranda-Filho, Ulisses Ramos Montarroyos, Celina Maria Turchi Martelli, Thalia Velho Barreto de Araújo, Laura C. Rodrigues.
- Investigation: Ricardo Arraes de Alencar Ximenes, Demócrito de Barros Miranda-Filho, Ulisses Ramos Montarroyos, Celina Maria Turchi Martelli, Thalia Velho Barreto de Araújo, Maria de Fátima Pessoa Militão de Albuquerque, Wayner Vieira Souza, Liana O. Ventura, Camila V. Ventura, Adriana L. Gois, Mariana C. Leal, Danielle Maria da Silva Oliveira, Sophie Helena Eickmann, Maria Durce C. G. Carvalho, Paula F. S. da Silva, Maria Angela Wanderley Rocha, Regina Coeli Ferreira Ramos, Sinval Pinto Brandão-Filho, Marli Tenorio Cordeiro, Luciana Caroline Albuquerque Bezerra, George Dimech, Sandra Valongueiro, Pedro Pires, Priscila Mayrelle da Silva Castanha, Rafael Dhalia, Ernesto Torres Azevedo Marques-Júnior, Laura C. Rodrigues.

- Methodology: Ricardo Arraes de Alencar Ximenes, Demócrito de Barros Miranda-Filho, Ulisses Ramos Montarroyos, Celina Maria Turchi Martelli, Thalia Velho Barreto de Araújo, Laura C. Rodrigues.
- **Project administration:** Ricardo Arraes de Alencar Ximenes, Demócrito de Barros Miranda-Filho, Ulisses Ramos Montarroyos, Celina Maria Turchi Martelli, Thalia Velho Barreto de Araújo.
- **Resources:** Ricardo Arraes de Alencar Ximenes, Demócrito de Barros Miranda-Filho, Ulisses Ramos Montarroyos, Celina Maria Turchi Martelli, Thalia Velho Barreto de Araújo, Laura C. Rodrigues.
- Supervision: Ricardo Arraes de Alencar Ximenes, Demócrito de Barros Miranda-Filho, Ulisses Ramos Montarroyos, Celina Maria Turchi Martelli, Thalia Velho Barreto de Araújo, Laura C. Rodrigues.
- Visualization: Ricardo Arraes de Alencar Ximenes, Demócrito de Barros Miranda-Filho, Ulisses Ramos Montarroyos, Celina Maria Turchi Martelli, Thalia Velho Barreto de Araújo, Laura C. Rodrigues.
- Writing original draft: Ricardo Arraes de Alencar Ximenes, Demócrito de Barros Miranda-Filho, Ulisses Ramos Montarroyos, Celina Maria Turchi Martelli, Thalia Velho Barreto de Araújo, Elizabeth Brickley, Maria de Fátima Pessoa Militão de Albuquerque, Wayner Vieira Souza, Liana O. Ventura, Camila V. Ventura, Adriana L. Gois, Mariana C. Leal, Danielle Maria da Silva Oliveira, Sophie Helena Eickmann, Maria Durce C. G. Carvalho, Paula F. S. da Silva, Maria Angela Wanderley Rocha, Regina Coeli Ferreira Ramos, Sinval Pinto Brandão-Filho, Marli Tenorio Cordeiro, Luciana Caroline Albuquerque Bezerra, George Dimech, Sandra Valongueiro, Pedro Pires, Priscila Mayrelle da Silva Castanha, Rafael Dhalia, Ernesto Torres Azevedo Marques-Júnior, Laura C. Rodrigues.
- Writing review & editing: Ricardo Arraes de Alencar Ximenes, Demócrito de Barros Miranda-Filho, Ulisses Ramos Montarroyos, Celina Maria Turchi Martelli, Thalia Velho Barreto de Araújo, Elizabeth Brickley, Maria de Fátima Pessoa Militão de Albuquerque, Wayner Vieira Souza, Liana O. Ventura, Camila V. Ventura, Adriana L. Gois, Mariana C. Leal, Danielle Maria da Silva Oliveira, Sophie Helena Eickmann, Maria Durce C. G. Carvalho, Paula F. S. da Silva, Maria Angela Wanderley Rocha, Regina Coeli Ferreira Ramos, Sinval Pinto Brandão-Filho, Marli Tenorio Cordeiro, Luciana Caroline Albuquerque Bezerra, George Dimech, Sandra Valongueiro, Pedro Pires, Priscila Mayrelle da Silva Castanha, Rafael Dhalia, Ernesto Torres Azevedo Marques-Júnior, Laura C. Rodrigues.

References

- Microcephaly Epidemic Research Group. Microcephaly in Infants, Pernambuco State, Brazil, 2015. Emerg Infect Dis. 2016; 22(6):1090–3. Available from: https://doi.org/10.3201/eid2206.160062 PMID: 27071041
- Dine MS, Gartside PS, Glueck CJ, Rheins L, Greene G, Khoury P. Relationship of Head Circumference to Length in the First 400 Days of Life: A Mnemonic. Pediatrics. 1981; 67(4):506–8. Available from: https://pediatrics.aappublications.org/content/67/4/506. PMID: 7254972
- Vasconcelos RAL, Ximenes RAA, Calado AA, Martelli CMT, Gonçalves AV, Brickley EB, et al. Cryptorchidism in children with Zika-related microcephaly: case series of a new condition. Am J Trop Med Hyg. 2020; 102(5): 982–4. https://doi.org/10.4269/ajtmh.19-0753 PMID: 32157994
- Moore CA, Staples JE, Dobyns WB, Pessoa A, Ventura CV, Fonseca EB, et al. Characterizing the Pattern of Anomalies in Congenital Zika Syndrome for Pediatric Clinicians. JAMA Pediatr. 2017; 171 (3):288–95. Avaliable from: https://jamanetwork.com/journals/jamapediatrics/fullarticle/2579543. https://doi.org/10.1001/jamapediatrics.2016.3982 PMID: 27812690

- Miranda-Filho DB, Martelli CMT, Ximenes RAA, Araújo TVB, Rocha MAW, Ramos RCF, et al. Initial Description of the Presumed Congenital Zika Syndrome. Public Health. 2016; 106(4):598–600. Available from: https://doi.org/10.2105/AJPH.2016.303115 PMID: 26959258
- Brasil P, Pereira Júnior JP, Moreira ME, Ribeiro-Nogueira RM, Damasceno L, Wakimoto M, et al. Zika Virus Infection in Pregnant Women in Rio de Janeiro. N Engl J Med. 2016; 375(24):2321–34. Available from: https://doi.org/10.1056/NEJMoa1602412 PMID: 26943629
- Hoen B, Schaub B, Funk AL, Ardillon V, Boullard M, Cabié A, et al. Pregnancy outcomes after ZIKV infection in French territories in the Americas. N Engl J Med. 2018; 378(11):985–94. Available from: https://doi.org/10.1056/NEJMoa1709481 PMID: 29539287
- Reynolds MR, Jones AM, Petersen EE, Lee EH, Rice ME, Bingham A, et al. Update on zika virus–associated birth defects and evaluation of all U.S. infants with congenital zika virus exposure—U.S. zika pregnancy registry, 2016. MMWR Morb Mortal Wkly Rep. 2017; 66(13):366–73. Available from: https://doi.org/10.15585/mmwr.mm6613e1 PMID: 28384133
- Shapiro-Mendoza CK, Rice ME, Galang RR, Fulton AC, VanMaldeghem K, Prado MV, et al. Pregnancy outcomes after maternal Zika virus infection during pregnancy—U.S. territories, January 1, 2016–April 25, 2017. MMWR Morb Mortal Wkly Rep. 2017; 66(23):615–21. Available from: https://doi.org/10. 15585/mmwr.mm6623e1 PMID: 28617773
- Rice ME, Galang RR, Roth NM, Ellington SR, Moore CA, Valencia-Prado M, et al. Zika-associated birth defects and neurodevelopmental abnormalities possibly associated with congenital zika virus infection —U.S. Territories and freely associated states, 2018. MMWR Morb Mortal Wkly Rep. 2018; 67 (31):858–67. Available from: https://doi.org/10.15585/mmwr.mm6731e1 PMID: 30091967
- Honein MA, Dawson AL, Petersen EE, Jones AM, Lee EH, Yazdy MM, et al. Birth defects among fetuses and infants of US women with evidence of possible Zika virus infection during pregnancy. JAMA. 2017; 317(1):59–68. Available from: https://doi.org/10.1001/jama.2016.19006 PMID: 27960197
- Pomar L, Vouga M, Lambert V, Pomar C, Hcini N, Jolivet A, et al. Maternal-fetal transmission and adverse perinatal outcomes in pregnant women infected with Zika virus: Prospective cohort study in French Guiana. BMJ. 2018; 363:k4431. Available from: https://doi.org/10.1136/bmj.k4431 PMID: 30381296
- Nogueira ML, Nery Júnior NRR, Estofolete CF, Terzian ACB, Guimarães GF, Zini N, et al. Adverse birth outcomes associated with Zika virus exposure during pregnancy in São José do Rio Preto, Brazil. Clin Microbiol Infect. 2018; 24(6):646–52. Available from: <u>https://doi.org/10.1016/j.cmi.2017.11.004</u> PMID: 29133154
- 14. WHO. WHO statement on the first meeting of the International Health Regulations (2005) (IHR 2005) Emergency Committee on Zika virus and observed increase in neurological disorders and neonatal malformations. Geneva: World Health Organization, 2016 Feb 1 [Cited 2019 Nov 8]. Available from: https://www.who.int/news-room/detail/01-02-2016-who-statement-on-the-first-meeting-of-theinternational-health-regulations-(2005)-(ihr-2005)-emergency-committee-on-zika-virus-and-observedincrease-in-neurological-disorders-and-neonatal-malformations.
- Araújo TVB, Rodrigues LC, Ximenes RAA, Miranda-Filho DB, Montarroyos UR, Melo APL, Valongueiro S, et al. Association between Zika virus infection and microcephaly in Brazil, January to May, 2016: preliminary report of a case-control study. Lancet Infect Dis. 2016; 16:1356–63. Available from: https://doi. org/10.1016/S1473-3099(16)30318-8 PMID: 27641777
- Araújo TVB, Ximenes RAA, Miranda-Filho DB, Souza WV, Montarroyos UR, Melo APL, et al. Association between microcephaly, Zika virus infection, and other risk factors in Brazil: Final report of a casecontrol study. Lancet Infect Dis. 2018; 18:328–36. Available from: https://doi.org/10.1016/S1473-3099 (17)30727-2 PMID: 29242091
- Paiva GFS, Silva DBN, Feijó CA. Exploratory note on onsumption and Socioeconomic classification in Brazil based on evidences from the family expenditure survey. Rev Econ Contemp. 2016; 20(2):207– 28. Available from: https://dx.doi.org/10.1590/198055272021.
- Ximenes RAA, Miranda-Filho DB, Brickley EB, Montarroyos UR, Martelli CMT, Araújo TVB, et al. Zika virus infection in pregnancy: Establishing a case definition for clinical research on pregnant women with rash in an active transmission setting. PLoS Negl Trop Dis. 2019; 13(10):e0007763. Available from: https://journals.plos.org/plosntds/article?id=10.1371/journal.pntd.0007763. https://doi.org/10.1371/journal.pntd.0007763 PMID: 31589611
- Lanciotti RS, Kosoy OL, Laven JJ, Velez JO, Lambert AJ, Johnson AJ, et al. Genetic and serologic properties of Zika virus associated with an epidemic, Yap State, Micronesia, 2007. Emerg Infect Dis. 2008; 14(8):1232–9. Available from: https://doi.org/10.3201/eid1408.080287.
- CDC. Zika MAC-ELISA: Instructions for use. Atlanta: Centers for Disease Control and Prevention. 2018 Feb 1 [Cited, 2020 Jan 15]. Available from: https://www.cdc.gov/zika/pdfs/zika-mac-elisa-instructionsfor-use.pdf.

- Nascimento EJM, Huleatt JW, Cordeiro MT, Castanha PMS, George JK, Grebe E, et al. Development of antibody biomarkers of long term and recent dengue virus infections. J Virol Methods. 2018; 257:62– 8. Available from: https://doi.org/10.1016/j.jviromet.2018.04.009 PMID: 29684416
- Castanha PMS, Braga C, Cordeiro MT, Souza AI, Silva CD, Martelli CMT, et al. Placental Transfer of Dengue Virus (DENV)–Specific Antibodies and Kinetics of DENV Infection–Enhancing Activity in Brazilian Infants. J Infect Dis. 2016; 214:265–72. Available from: <u>https://doi.org/10.1093/infdis/jiw143</u> PMID: 27056951
- Cordeiro MT, Brito CAA, Pena LJ, Castanha PMS, Gil LHVG, Lopes KGS, et al. Zika virus IgM-specific based diagnostic is highly correlated with detection of neutralising antibodies in neonates with congenital disease. LSHTM Res Online. 2016; 241(12):1897–904. Available from: https://doi.org/10.1093/ infdis/jiw477.
- 24. WHO. Child Growth Standards: Head circumference-for-age, arm circumference-for-age, triceps skinfold-for-age and subscapular skinfold-for-age: Methods and development. Geneva: World Health Organization, 2007 [Cited 2020 Jan 15]. Available from: https://apps.who.int/iris/handle/10665/43706.
- Villar J, Giuliani F, Bhutta ZA, Bertino E, Ohuma EO, Ismail LC, et al. Postnatal growth standards for preterm infants: The Preterm Postnatal Follow-up Study of the INTERGROWTH-21stProject. Lancet Glob Heal. 2015; 3 (11):e681–91. Available from: https://doi.org/10.1016/S2214-109X(15)00163-1.
- Almeida FCF. Protocolo de avaliação clínica da disfagia pediátrica (PAD-PED). Monography. Universidade de São Paulo. 2014. Available from: https://repositorio.usp.br/item/002649232.
- Joint Committee on Infant Hearing. Year 2007 Position Statement: Principles and Guidelines for Early Hearing Detection and Intervention Programs. Pediatrics. 2007; 120(4):898–921. Available from: https://doi.org/10.1542/peds.2007-2333 PMID: 17908777
- Ashwal S, Michelson D, Plawner L, Dobyns WB. Practice Parameter: Evaluation of the child with microcephaly (an evidence-based review). Neurology. 2009; 73(11):887–97. Available from: <u>http://www.neurology.org/cgi/doi/10.1212/WNL.0b013e3181b783f7</u>. PMID: 19752457
- 29. Brazilian Ministry of Health [Ministério da Saúde—MS]. Ministério da Saúde atualiza números de microcefalia relacionados ao Zika. 2015 Dec 15.[cited 2020 Aug 14]. In: Brazilian Government Portal [Portal do Governo Brasileiro]. Distrito Federal: MS. Available from: https://www.saude.gov.br/noticias/ agencia-saude/21254-ministerio-da-saude-atualiza-numeros-de-microcefalia-relacionados-ao-zika.
- Hazin AN, Poretti A, Martelli CMT, Huisman TA. Computed tomographic findings in microcephaly associated with Zika Virus. N Engl J Med. 2016; 374(22):2193–5. Available from: <u>https://doi.org/10.1056/</u> NEJMc1603617 PMID: 27050112
- Ventura LO, Ventura CV., Lawrence L, Van der Linden V, Van der Linden A, Gois AL, et al. Visual impairment in children with congenital Zika syndrome. J AAPOS. 2017; 21(4):295–9. Available from: https://doi.org/10.1016/j.jaapos.2017.04.003 PMID: 28450178
- Ventura CV, Maia M, Dias N, Ventura LO, Belfort R. Zika: Neurological and ocular findings in infant without microcephaly. Lancet. 2016; 387:2502. Available from: <u>https://doi.org/10.1016/S0140-6736(16)</u> 30776-0 PMID: 27287830
- Leal MC, Muniz LF, Ferreira TSA, Santos CM, Almeida LC, Van der Linden V, et al. Hearing loss in infants with microcephaly and evidence of congenital zika virus infection—Brazil, November 2015–May 2016. MMWR Morb Mortal Wkly Rep. 2016; 65(34):917–9. Available from: https://doi.org/10.15585/ mmwr.mm6534e3 PMID: 27585248
- Leal MC, Van der Linden V, Bezerra TP, Valois L de, Borges ACG, Antunes MMC, et al. Characteristics of dysphagia in infants with microcephaly caused by congenital zika virus infection, Brazil, 2015. Emerg Infect Dis. 2017; 23(8):1253–9. Available from: <u>https://doi.org/10.3201/eid2308.170354</u> PMID: 28604336
- Khan K, Khan W. Congenital toxoplasmosis: An overview of the neurological and ocular manifestations. Parasitol Int. 2018; 67(6):715–21. Available from: <u>https://doi.org/10.1016/j.parint.2018.07.004</u> PMID: 30041005
- Enders G, Daiminger A, Bäder U, Exler S, Enders M. Intrauterine transmission and clinical outcome of 248 pregnancies with primary cytomegalovirus infection in relation to gestational age. J Clin Virol. 2011; 52(3):244–6. Available from: https://doi.org/10.1016/j.jcv.2011.07.005 PMID: 21820954
- Souza WV, Albuquerque M de FPM, Vazquez E, Bezerra LCA, Mendes A da CG, Lyra TM, et al. Microcephaly epidemic related to the Zika virus and living conditions in Recife, Northeast Brazil. BMC Public Health. 2018; 18:130. https://doi.org/10.1186/s12889-018-5039-z PMID: 29329574
- Ximenes RAA, Albuquerque M de FPM, Souza WV, Montarroyos UR, Diniz GTN, Luna CF, Rodrigues LC. Is it better to be rich in a poor area or poor in a rich area? A multilevel analysis of a case-control study of social determinants of tuberculosis. Int J Epidemiol. 2009; 38:1285–96. https://doi.org/10.1093/ ije/dyp224 PMID: 19656772

- 39. Braga C, Ximenes RA, Albuquerque M, Souza WV, Miranda J, Brayner F, et al. Evaluation of a social and environmental indicator used in the identification of lymphatic filariasis transmission in urban centers. Cad Saúde Pública. 2001; 17(5):1211–8. <u>https://doi.org/10.1590/s0102-311x2001000500021</u> PMID: 11679895
- 40. Ferreira MLB, Albuquerque M de FPM, Brito CAA, França RFO, Moreira AJP, Machado MIM, et al. Neurological disease in adults with Zika and chikungunya virus infection in Northeast Brazil: a prospective observational study. Lancet Neurol. 2020; 19(10):826–39.
- **41.** Paixao ES, Leong W-Y, Rodrigues LC, Wilder-Smith A. Asymptomatic Prenatal Zika Virus Infection and Congenital Zika Syndrome. Open Forum Infect Dis. 2018; 5(4):ofy073. Available from: <u>https://doi.org/10.1093/ofid/ofy073 PMID: 29732381</u>